Institut for Matematik og Datalogi Syddansk Universitet

May 15, 2012 JFB DM830 Network Security – F12 – Lecture 8

Lecture, May 9

We finished up through section 7.2 of chapter 7.

Lecture, May 14

We will finish chapter 7 and begin on chapter 8.

Lecture, May 18

We will finish chapter 8.

Announcement

In addition to possibilities for being “instruktor” in a course, there are positions as advisor for first year study groups for the fall. See

http://www.jobs.sdu.dk/vis stilling.php?id=7288&lang=da

Discussion section: May 21

There will be a presentation on DDOS, concentrating on defenses. Do the following problems:

1. Problem 8.1 in the textbook.

2. What does “mutable” mean? Name some fields of an IP header which are mutable and some which are not. How is this relevant in IPsec?

3. Problem 8.5 in the textbook.

4. Problem 8.6 in the textbook.

5. Do the following Review Questions: 10.1, 10.2, 10.4. 6. Problem 10.2 in the textbook. 7. Problem 10.4 in the textbook. 8. Problem 10.6 in the textbook.